Author’s response to reviews

Title: An outline of anemia among adolescent girls in Bangladesh: findings from a cross-sectional study

Authors:

Sabuj Kanti Mistry (sabuj.km@brac.net)
Fatema Tuz Jhohura (fatema.jhohura@brac.net)
Fouzia Khanam (fouzia.khanam@brac.net)
Fahmida Akter (fahmida.akter@brac.net)
Safayet Khan (safayet.k@brac.net)
Fakir Md Yunus (fakir.yunus@usask.ca)
Md. Belal Hossain (belal.h@brac.net)
Kaosar Afsana (afsana.k@brac.net)
Raisul Haque (raisul.h@brac.net)
Mahfuzar Rahman (mahfuzar.rahman@brac.net)

Version: 3 Date: 22 Jul 2017

Author’s response to reviews:

Editor Comments:

Please correct the decimal point in line 236-237 page 12:

- The text has been revised: “It was also noted that mean hemoglobin level was 11.82 ± 1.15 g/dl, while it was 10.99 ± 0.9 g/dl and 12.7 ±0.61 g/dl respectively.....” (Result section, line 236-37, page 12)